Author’s response to reviews

Title: Elevated serum levels of aminotransferases in relation to unhealthy foods intake: Tehran Lipid and Glucose Study

Authors:

Parvin Mirmiran (mirmiran@endocrine.ac.ir)
Zahra Gaeini (zahragn1992@gmail.com)
Zahra Bahadoran (zahrabahadoran@yahoo.com;z.bahadoran@endocrine.ac.ir)
Fereidoun Azizi (azizi@endocrine.ac.ir)

Version: 4 Date: 28 Sep 2019

Author’s response to reviews:

C1. Please rename the "Introduction" section to "Background".
R: thanks the editor for good points, agreed and revised (page 4, line 1)

C2. The Availability of data and materials section refers to the raw data used in your study. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):
R: Agreed and added in the revised manuscript (page 13, lines 17-18)

C3. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
R: Agreed and revised (page 14, lines 2-4)

C4. As Zahra Bahadoran is a member of the editorial board of BMC Endocrine Disorders please declare this in the Competing Interests section of the Declarations, so as to ensure transparency.
R: Agreed and added in the revised manuscript (13, line 19)

C5. Table 4 is not cited in the body of the manuscript. Please amend this.
R: Table 4 is cited at the page 9, lines 18-19 of the manuscript